



## Article Excel Methods to Design and Validate in Microelectronics (Complementary Metal–Oxide–Semiconductor, CMOS) for Biomedical Instrumentation Application

Graciano Dieck-Assad <sup>1</sup>, José Manuel Rodríguez-Delgado <sup>1</sup> and Omar Israel González Peña <sup>1,2,\*</sup>

- <sup>1</sup> Tecnológico de Monterrey, School of Engineering and Science, Av. Eugenio Garza Sada Sur No. 2501, col. Tecnológico, Monterrey 64849, Mexico; graciano.dieck.assad@tec.mx (G.D.-A.); jmrd@tec.mx (J.M.R.-D.)
- <sup>2</sup> Tecnológico de Monterrey, Institute for the Future of Education, Av. Eugenio Garza Sada Sur No. 2501, col. Tecnológico, Monterrey 64849, Mexico
- \* Correspondence: ogonzalez.pena@gmail.mx

Abstract: CMOS microelectronics design has evolved tremendously during the last two decades. The evolution of CMOS devices to short channel designs where the feature size is below 1000 nm brings a great deal of uncertainty in the way the microelectronics design cycle is completed. After the conceptual idea, developing a thinking model to understand the operation of the device requires a good "ballpark" evaluation of transistor sizes, decision making, and assumptions to fulfill the specifications. This design process has iterations to meet specifications that exceed in number of the available degrees of freedom to maneuver the design. Once the thinking model is developed, the simulation validation follows to test if the design has a good possibility of delivering a successful prototype. If the simulation provides a good match between specifications and results, then the layout is developed. This paper shows a useful open science strategy, using the Excel software, to develop CMOS microelectronics hand calculations to verify a design, before performing the computer simulation and layout of CMOS analog integrated circuits. The full methodology is described to develop designs of passive components, as well as CMOS amplifiers. The methods are used in teaching CMOS microelectronics to students of electronic engineering with industrial partner participation. This paper describes an exhaustive example of a low-voltage operational transconductance amplifier (OTA) design which is used to design an instrumentation amplifier. Finally, a test is performed using this instrumentation amplifier to implement a front-end signal conditioning device for CMOS-MEMS biomedical applications.

**Keywords:** freeware; open science; analog microelectronics design; long channel transistors; short channel transistors; integrated circuit design; CMOS design; VLSI; higher education; educational innovation; integrated circuit layout; complex thinking

## 1. Introduction

Currently, we have high-capacity technology of analog and digital electronic devices due to the micro-components that are increasingly becoming smaller in scale. In fact, in 1965, Gordon E. Moore predicted (Moore's law) that every 2 years the number of transistors in a microprocessor would double. This law worked for the first 10 years [1]; then it became a joke between engineers who said that "now people predict that the end of Moore's law doubles every 2 years." Since then, transistor integration has been observed as illustrated in Figure 1, where Moore's law was in force for a long time. However, the limits towards a scale in nanotechnology have shown that although Moore's law no longer applies, significant efforts are still being made to continue reducing the size of the micro and nano components. For example, recently design and process in semiconductors have been done with the development of the world's first chip with 2 nanometers (nm) nanosheet technology [2].



Citation: Dieck-Assad, G.; Rodríguez-Delgado, J.M.; González Peña, O.I. Excel Methods to Design and Validate in Microelectronics (Complementary Metal–Oxide– Semiconductor, CMOS) for Biomedical Instrumentation Application. *Sensors* **2021**, *21*, 7486. https://doi.org/10.3390/s21227486

Academic Editors: Jong-Ryul Yang and Seong-Tae Han

Received: 25 September 2021 Accepted: 30 October 2021 Published: 11 November 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).



**Figure 1.** Behavior of the technological development for the small-scale design of transistors on microchips doubling every two years (Moore's law) [1].

Therefore, with this context in which the microchip technology is reducing in size over time, college students with different science and engineering majors need to understand the "know-how" of designing circuits that integrate microchips. Unfortunately, often this knowledge is taught with specialized complex (and expensive) software, which is sometimes not readily available in many universities around the world.

Analog and digital integrated circuit design has emphasized a good understanding of analog and digital electronic circuits to model, simplify, analyze, and simulate microelectronic devices before developing layouts and sending devices to IC fabrication foundry consortiums. Design engineers find SPICE simulations useful to validate "thinking models" [3,4] to verify cause–effect relationships and to ensure that assumptions made are appropriate in the design process. Microelectronics courses and specialized workshops (and webinars) teach conceptual design where the following concepts are emphasized.

- The importance of the integrated circuit (IC) design in microelectronics engineering is embedded into electronic engineering innovation and design flow.
- Analog CMOS microelectronics focus on the electrical and physical design processes.

IC design is important in device development for microelectronics engineering innovation and is embedded into the design flow which may have continuous iterations to optimize designs by decision refinements. In the process of CMOS microelectronics conceptual design [5–7], the electrical device specification requires active and passive models for creating, verifying, and determining the robustness of the design. This process involves the selection of a conceptual circuit, the analysis of the selected circuit, the possibility of a modification to the circuit, and the verification of the circuit solution. The physical electronics design process [8–10] consists of representing the electrical device in a 2D layout consisting of many different geometrical rectangles at various levels (layers). This layout is then used to implement a 3D integrated circuit during the fabrication process. The device conceptual model follows a process to obtain a layout. This process includes [3,4]:

- 1. The W/L (width/length of transistors) values and schematic (usually from SPICE simulations).
- 2. A computer-aided tool (CAD) system is used to enter geometries.
- 3. The engineer must obey a set of rules called design rules. These rules establish the fabrication limitations and ensure that the device is robust and reliable.
- 4. Once the layout is complete, a process called layout versus schematic (LVS) is applied to determine if the physical layout represents the electrical schematic.
- 5. Parasitic elements are extracted, once the physical dimensions of the designs are known. They include capacitances from conductor to ground or between conductors and bulk resistances.
- 6. The parasitic elements are entered into the simulation database and the design is re-stimulated to ensure they will not cause a device failure.



This process is depicted in the flow diagram shown below in Figure 2.

Figure 2. Block diagram showing the design procedure to reach the Layout conceptual model.

Once the first approach to design is terminated, the process continues with design testing which consists of coordinating, planning, and implementing the measurement of the integrated circuit performance. The objective is to compare the experimental performance with the specifications and/or simulation results. Several tests are available, e.g., functional: verification of the nominal specifications; parametric: verification of the characteristics to within a specified tolerance, verification of the static (AC and DC) characteristics of the circuit or system, and verification of the dynamic (transient) characteristics of a circuit or system. Additional testing could include device testing performed at the wafer level or package level and detailed testing that removes the influence of measurement system in the device performance.

The conceptual design of analog integrated circuits in new devices is now shaped by rules such as: consumers generate a need for new integrated circuits, design engineers have an open possibility of participating in designs, time to develop a product is reduced, profit in products and prototypes are not readily necessary, and the new concept of "crowd designing" [3] plays an important role in device development. This article discusses Excel methods to perform "paper and pencil" calculations ("thinking model") in the conceptual design of CMOS analog integrated circuits which are validated using ELECTRIC-LTSpice to

provide the initial characteristics of the device. Several methodologies have been developed for CMOS device design and testing but most of them are very specific to the final prototype or experimental application [11–17]. The Excel methods analyzed in this study focus on providing didactic instruction in microelectronics for undergraduate and graduate students. Therefore, it seeks to focus learning by referring to world trends in conducting open science, providing the social appropriation of knowledge. The forefront is didactic management whose central axis is the catalyst of open innovation processes that have proven to be very successful disruptive models in open laboratories, universities, research centers, industry, and government for the development of emerging economies and public policies [18–26].

As a result, this study describes a method that uses an Excel spreadsheet to start the conceptual design from the point of view of the "thinking model" or the first cut evaluation of the design with "paper and pencil". In addition, several contributions stand out:

The Excel methods discussed here focus on microelectronics education for undergraduate and graduate students. The article describes a simple method using an Excel spreadsheet to initiate conceptual design from the standpoint of "thinking model" or "paper and pencil" first cut evaluation of the design. Furthermore, several contributions are emphasized:

- (a) This work describes both the conceptual design evaluations using the traditional equations and prepares the way for the layout implementation by setting up the schematic of the design.
- (b) In addition, this study describes that once the schematic simulation agrees with the scheduled specifications and the "thinking model" provides a possible design solution, the layout is developed and a comparison between the output results with the thinking model specs validates the third phase of the process.
- (c) Moreover, the methodology applied in this study can be developed for more complicated CMOS analog integrated circuits (IC) conceptual designs, specifically for teaching purposes.
- (d) Complex conceptual designs can also be addressed using simple and readily available software (freeware) to teach with an open science view.
- (e) Furthermore, this article illustrates an exhaustive example of a low voltage operational transconductance amplifier (OTA) design for portable biomedical applications. The test is performed using this instrumentation amplifier to implement a front-end signal conditioning device for CMOS-MEMS biomedical applications.

This manuscript is organized as follows. Section 2 briefly describes device modeling to represent transistors using first principle equations that predict their behavior in different regimes and operating conditions. Section 3 provides the basic MOSFET modeling equations, starting with the threshold voltage calculations, the transconductance equations for the ohmic (sub-saturation), active (saturation), and subthreshold conditions, both for long channel and short channel devices. Section 3 also gives the typical MOSFET parameters for 0.5 mm CMOS technology, the small-signal model parameter equations, and useful resistance and capacitance calculations using the foundry process parameters for the technology. Section 4 describes the Excel methods that students use to develop their first cut approximations in conceptual designs of CMOS devices and before testing schematic simulations and layouts. This section explains three Excel methodologies: single straight, tabular straight, and two-dimensional processing methods to perform the evaluation of the device's conceptual design. This section also discusses resistance, capacitance, and differential amplifier conceptual designs as specific examples to apply the Excel methods. Section 5 discusses the Excel methods applied for complete amplifier design. Here the cascode amplifier and the OTA (Operational Transconductance Amplifier) are used as examples of how students use the basic two-dimensional Excel methodology to solve CMOS microelectronic conceptual design devices. Section 6 illustrates a complete case study, developed by students and wrapped up by their instructor, of a low-power high-gain operational amplifier for biomedical applications. In this section, comparison of the design is performed with respect to a particular device appearing in the technical literature and

an application of a CMOS-MEMS signal conditioning is developed considering the requirement of conditioning a differential mode temperature sensor's output to obtain a readily available low voltage representation of the temperature of a micro hotplate multisensor platform. Finally, Section 7 wraps up the paper with illustrative conclusions about how these Excel methods have provided extraordinary insights to undergraduate students in their effort to consolidate a good understanding of microelectronics conceptual integrated circuit (IC) design.

## 2. Materials and Methods—Device Modeling

The process of device modeling consists of representing the electrical properties of devices using mathematical equations, circuits, graphs, correlations, and energy conservation laws. Models allow predictions and validation of circuit performance with uncertainties coming from no-idealities and non-linear behaviors in electronic components. Typical equations and conservation laws are Ohm's law, large- and small-signal models of MOS-FET transistors, VTC curves, and I-V curves of diodes. The final goals are to simplify the cause–effect relationships allowing the engineer to understand and consider decisions that increase performance in the circuit.

Analog integrated circuits in microelectronic conceptual design are developed using a non-hierarchical structure where the use of repeated blocks is only possible in few devices, and therefore the design process is complex and challenging. To handle this, design engineers use hierarchy whenever possible, use good organization techniques, efficiently document the design, provide reasonable and reliable assumptions and simplifications, and eventually validate the conceptual designs using simulation experiments. Assumptions and simplifications are used to emphasize the essential characteristics by neglecting the non-dominant effects in the design. The challenge of teaching microelectronics is to develop an insight into the design process without requiring specialized professional software which is not readily available in many universities around the world. Figure 3 shows the microelectronics design process, on the left side, with the insight given on the right side, using Excel methods and other simulation techniques which are readily available to universities.



**Figure 3.** The CMOS microelectronics design process and the insight provided by the Excel methods in the quest to teach microelectronics design with simple tools.

Analog integrated circuit design and device evaluation have reached a level of maturity in established applications such as digital to analog and analog to digital conversion systems, front end signal conditioning devices, instrumentation channel devices, bandgap reference sources, DC to DC power conversion drives, and other important microelectronic circuits [3]. Finally, analog circuit conceptual designs have significant applications in devices where speed and power have an overwhelming advantage over digital devices. This paper reviews the long and short channel models for CMOS devices and further application of long channel equations using Excel methods for first cut approximations before computer simulations and layout development. Those approximation models can be found in many CMOS microelectronics specialized textbooks [3,4,7,27–29] and they are summarized here for completeness in this discussion.

### 3. Results—Transistor Models for Analog Conceptual Design

### 3.1. Threshold Voltage Calculation

The evaluation of threshold voltage,  $V_T$ , is fundamental in the development of CMOS microtechnology because gives the necessary condition for allowing operation of the transistors in the right zone.

$$V_T = [V_{T0}] + \gamma \left( \sqrt{|2\varphi_F| + V_{SB}} - \sqrt{|2\varphi_F|} \right)$$
(1)

where  $\gamma$  is the body factor,  $2\varphi_F$  is the Fermi potential,  $V_{T0}$  is the zero bias threshold voltage, and  $V_{SB}$  is the potential difference between source and bulk of the device.

### 3.2. Current Equations for Long Channel Devices

The drain-source current equations for the MOSFET long channel device are given below for  $V_{GS} \ge V_T$  and  $k = \mu C_{ox}(W/L)$ :

Saturation zone:  $V_{DS} \ge V_{GS} - V_T$ 

$$T_D = \frac{k}{2} [V_{GS} - V_T]^2 (1 + \lambda V_{DS})$$
 (2)

Ohmic zone:  $V_{DS} \leq V_{GS} - V_T$ 

$$i_D = \frac{\kappa}{2} [2(V_{GS} - V_T) - V_{DS}] V_{DS}$$
(3)

For  $V_{GS} \leq V_T$  the transistor plays in the subthreshold zone with the following exponential behavior, like the bipolar junction transistor:

$$i_D = I_t \frac{W}{L} e^{\frac{(V_{GS} - V_T)}{nV_t}} \tag{4}$$

where the nomenclature and parameters of the model are as follows:  $i_D$  is the drain to source current flowing in the MOSFET,  $\mu$  is the charge carrier mobility,  $C_{ox}$  is the SiO<sub>2</sub> oxide capacitance, (*W/L*) is the width/length transistor ratio,  $V_{GS}$  is the gate to source potential differences,  $V_{DS}$  is the drain to source potential differences,  $\lambda$  is the transistor 's channel modulation parameter,  $I_t$  is the subthreshold saturation current (<1  $\mu$ A),  $V_t$  is the thermal voltage (~26 mV at 25 °C), and n is a subthreshold constant (~1 to 2).

#### 3.3. Velocity Saturation and Effective Mobility in Current Equations of Short Channel Devices

To consider velocity saturation and effective mobility in short channel MOSFETs adjustments are made in the model equations presented above. This is particularly important in the saturated and ohmic regions of operation. The drain-source current equations for the MOSFET short channel device are given below for  $V_{GS} \ge V_T$ , having saturation velocity  $v_{sat} = \mu_e E_c/2$ , where  $\mu_e$  is now the effective mobility of charge carriers and  $E_c$  is the critical field for which the carrier saturation occurs. Saturation zone:

$$V_{DS}(sat) = \frac{(V_{GS} - V_T)E_cL}{(V_{GS} - V_T) + E_cL}$$
(5)

$$i_D = \frac{W\mu_e C_{ox} E_c}{2} \left[ \frac{(V_{GS} - V_T)^2}{(V_{GS} - V_T) + E_c L} \right]$$
(6)

Ohmic zone:

$$V_{DS}(ohmic - limit) = V_{DS}(sat) = \frac{(V_{GS} - V_T)E_cL}{(V_{GS} - V_T) + E_cL}$$
(7)

$$i_D = \frac{W\mu_e C_{ox} E_c}{(E_c L + V_{DS})} \left[ (V_{GS} - V_T) V_{DS} - \frac{V_{DS}}{2} \right] V_{DS}$$
(8)

If we summarize the SPICE-level I models of the MOSFET transistors to be used in conceptual CMOS designs, Table 1 describes the equations considering the ohmic and saturation regions in the so-called strong inversion regime.

Table 1. Summary of CMOS N-Channel transistor large signal model.

	Long Channel L > 1 $\mu$ m	Short Channel L < 1 $\mu$ m						
Ohmic	$i_{DS} = \frac{k'W}{L} \left[ (V_{GS} - V_T) - \frac{V_{DS}}{2} \right] V_{DS}$	$i_{DS} = rac{\mu_e C_{ox} E_c W}{(E_c L + V_{DS})} \left[ \left( V_{GS} - V_T \right) V_{DS} - rac{V_{DS}}{2} \right] V_{DS}$						
$V_{DS}(sat)$	$V_{DS}(sat) = V_{GS} - V_T$	$V_{DS(sat)} = \frac{(V_{GS} - V_T)E_cL}{(V_{GS} - V_T) + E_cL}$						
Saturation	$i_{DS} = \frac{kW}{2L} [V_{GS} - V_T]^2$	$i_D = \frac{W\mu_e C_{ox} E_c}{2} \left[ \frac{(V_{GS} - V_T)^2}{(V_{GS} - V_T) + E_c L} \right]$						
Notes:	k'	$=\mu_0 C_{ox}$						
	$E_c$ = critical horizontal field $E_y$							
	$\mu_e$ = carrier mobility considering horizontal field Ey							

Moreover, for "paper and pencil" calculations ("Thinking Modeling") the parameters used for 0.5  $\mu$ m CMOS technology are described in Table 2.

Table 2. MOSFET parameters for CMOS 0.5 µm process (C5\_models) obtained in ON-SEMI Wafer Runs [6].

Parameter Symbol Description		Parameter Valu	Parameter Value					
		N-Channel	P-Channel	Units				
$V_{T0}$	Threshold voltage ( $V_{bs} = 0$ )	0.76	-0.96	V				
Κ'	Transconductance Parameter in saturation	115.6	37.8	$\mu A/V^2$				
$\gamma$	Bulk threshold parameter	0.49	0.56	$V^{1/2}$				
λ	Channel Modulation parameter	0.04	0.05	$V^{-1}$				

## 3.4. Small-Signal Model of Devices

The useful small-signal equations for MOSFET transistors correspond to the saturation zone of the device. The two most important parameters are  $g_m$ , transconductance gain, and  $g_{ds}$ , output conductance of the device which provides its output resistance [3,4]. The equations for those parameters are described as follows.

$$g_m = \frac{di_{DS}}{dV_{GS}} \Big|_Q = \sqrt{2\beta I_D (1 + \lambda V_{DS})} \cong \sqrt{2\beta I_{DS}}$$
(9)

$$g_{ds} = \frac{di_{DS}}{dV_{DS}} \bigg|_{Q} = \frac{\lambda I_{DS}}{1 + \lambda V_{DS}} \cong \lambda I_{DS}$$
(10)

From those expressions,  $\lambda$  is the channel modulation parameter shown in Table 2 and  $\beta = k = \mu Cox(W/L)$  which was defined earlier for the large-signal model. These parameters are evaluated at the quiescent Q point where the device operates and the small-signal ignites.

### 3.5. Parasitic Capacitances

The most important parasitic elements of the MOSFET are the capacitances due to the inherent field-effect operation of this transistor. The oxide and p-n junction capacitances are responsible for limitations of the frequency response which generates poles and zeros which are frequently analyzed to determine the stability of the device and the dominant bandwidth of the system. Three capacitances are considered in the conceptual design modeling:

i. "Oxide capacitance" is formed by the SiO<sub>2</sub> between the gate terminal and the channel formed from drain to source. This capacitance is given by the gate oxide capacitance as follows.

$$C_g = WLC_{ox} = WL\epsilon_{ox}/t_{ox} \tag{11}$$

where  $\varepsilon_{ox}$  is the oxide dielectric constant and  $t_{ox}$  is the oxide thickness.

ii. "P-N junction capacitances" due to the reverse bias of the p-n junctions formed between drain/source to the substrate/bulk terminals of the device. This capacitance is evaluated using the built-in potential  $\varphi_B$ , built-in zero-bias junction capacitance  $C_{j0}$ , and the built-in junction capacitance  $C_j$  as follows.

$$\varphi_B = V_t ln\left(\frac{N_A N_D}{n_i^2}\right) \tag{12}$$

$$C_{j0} = \sqrt{\frac{\epsilon_{Si}qN_A}{2\varphi_B}} \tag{13}$$

$$C_j = \frac{C_{j0}}{\left(1 - \frac{V_j}{\varphi_B}\right)^m} \tag{14}$$

where  $n_i$  is the intrinsic carrier concentration in silicon,  $N_A/N_D$  are numbers of acceptor/donor atom concentrations per volume in material,  $\varepsilon_{Si}$  is the silicon dielectric constant, and  $V_i$  is the reverse bias voltage applied.

#### 3.6. Passive Components

In microelectronic conceptual design, the passive components provide additional versatility in the consolidation and refinement of many integrated circuits that require compensation, parasitic element adjustment, antenna integration, and other possible maneuvers. For the conceptual modeling using the Excel methodologies, this paper considers only resistor and capacitor implementations. To implement resistances, Table 3 shows the values for sheet and contact resistances in the C5 ON-SEMI process [15] described by wafer runs from the corresponding foundry.

Table 3. Resistance process parameters for CMOS 0.5 µm process used in ON-SEMI wafer runs [13].

Process Parameters	N+	P+	POLY	POLY2	POLY2 HR	N <sub>well</sub>	M1	M2	M3	UNITS
Sheet resistance Contact resistance	82.4 59.6	106.7 152.5	23.2 16	40.8 26	1076	808	0.09 0.84	0.09 0.84	0.05 0.82	$\Omega/sq$ $\Omega$

From Table 3, N+ and P+ are the n-type and p-type doped active silicon materials, respectively; POLY, POLY2, and POLY2\_HR are polysilicon, polysilicon-2, and polysilicon-

2-high-resistivity materials, respectively; N<sub>WELL</sub> corresponds to the n-type well (where P-Type transistors are located); M1, M2, and M3 are metal-1, metal-2, and metal-3 layers, respectively, where the component connections are implemented inside the chip.

To implement capacitances, Table 4 shows the values for area, fringe and overlap capacitances in C5 ON-SEMI process [15]. The area capacitances for substrate, N+ active, P+ active, POLY, POLY2, M1 and M2 have units of  $aF/\mu^2$ , while the fringe and overlap capacitances are given in units of  $aF/\mu$ . Parameters from table IV are very important in the compensation of analog integrated circuits that are used in instrumentation channels for processing signals coming from sensors and transducers.

Table 4. Capacitance process parameters for CMOS 0.5 µm process used in ON-SEMI wafer runs [15].

Capacitance Parameters	N+	P+	POLY	POLY2	M1	M2	M3	N <sub>well</sub>	UNITS
Area(substrate)	416	710	86	-	29	12	8	91	aF/µm <sup>2</sup>
Area(N+ active)	-	-	2456	-	-	-	-	-	aF/µm²
Area(P+ active)	-	-	2456	-	-	-	-	-	aF/µm <sup>2</sup>
Area(POLY)	-	-	-	922	64	16	9	-	aF/µm²
Area(POLY2)	-	-	-	-	58	-	-	-	aF/µm²
Area(M1)	-	-	-	-	-	32	12	-	aF/µm²
Area(M2)	-	-	-	-	-	-	32	-	aF/µm <sup>2</sup>
Fringe(substrate)	345	236	-	-	51	34	26	-	aF/µm²
Fringe(POLY)	-	-	-	-	70	39	28	-	aF/µm²
Fringe(M1)	-	-	-	-	-	49	33	-	aF/µm <sup>2</sup>
Fringe(M2)	-	-	-	-	-	-	55	-	aF/µm²
Overlap(N+active)	-	-	191	-	-	-	-	-	aF/µm <sup>2</sup>
Overlap(P+active)	-	-	234	-	-	-	-	-	aF/µm <sup>2</sup>

The equations to implement passive components are given as follows. Resistances made in N<sub>well</sub>, N+, P+, POLY, POLY2 and POLY2\_HR:

$$R_T = [Sheet Resistance Parameter] \times L/W$$
(15)

Capacitances made by POLY-POLY or POLY-Metal:

$$C_T = [Capacitance \ Parameter] \times WL \tag{16}$$

Now that the model equations have been presented, the following sections will discuss the methodology used to describe the conceptual model for the CMOS microelectronic circuit.

### 4. Excel Methods for CMOS Design

Conceptual design involves the use of parameters and constants to compute and size values of components and electrical variables [16,17]. However, sometimes the number of specifications given is larger than the number of degrees of freedom available to the designer trying to comply and fulfill them [3,4,27–29]. Therefore, in the design process, a series of decision points is necessary to iterate the processing flow before obtaining the final solution. Several trials are necessary and sometimes, intermediate simulations are required to analyze different alternative solutions. In this case, the use of Excel methods is convenient while advancing in the design process. There are three sorts of recommended methods:

- a. Single straight processing.
- b. Tabular straight processing.
- c. Two-dimensional processing.

### 4.1. Single Straight Processing Excel Method

In the single straight processing, the method proceeds in a horizontal fashion with the CMOS technology parameters in the first block of columns. Then, the processing flow continues sequentially, column by column, until the final desired calculation values are obtained. If several conditions are considered, then a row repetition is developed accordingly. Figure 4 shows single straight processing to design resistances using active p+, active n+, polysilicon and polysilicon-2 CMOS elements with their respective contact resistances and varying W and L sizes. Figures 5 and 6 show the CMOS 500 nm IC layouts of 3.5 K and 1.5 K resistors, obtained from Electric-VLSI. The artwork was developed once the Excel method was used to calculate the number of sheets or squares required by the conceptual design.

Resistar	nce para	meters	Contact	Resistance			Equivale	ent Resis	tance
Ohms per square			Ohms				p+	n+	
p+	n+		p+	n+	W	L	Req p+	Req n+	
106.7	82.4		152.5	59.6	3	95	3531.3	2668.9	
Ohms p	Ohms per square						Equivale	ent Resist	tance
Poly	Poly-2		Poly	Poly-2	W	L	Poly	Poly-2	
23.2	40.8		16	26	5	334	1565.8	2751.4	

**Processing Flow** 

Figure 4. Single straight method to design CMOS resistances.

Active p+ Resistor W = 3 L = 95 R= Rsheet x L/W = Rsheet x 95/3

Figure 5. CMOS IC layout active p+ resistor design, 3.5 K, from Electric-VLSI.



Figure 6. CMOS IC layout polysilicon resistor, 1.5 K, from Electric-VLSI.

4.2. Tabular Straight Processing Excel Method

In the tabular processing, the method proceeds as usual, in a horizontal fashion to evaluate the design instance with the CMOS technology parameters in the first block of columns. However, this evaluation is repeated for a multiple number of instances while varying one or two specifications or parameters. Then, the processing flow continues sequentially, row by row, until the final desired instance is terminated. Each instance considered will be evaluated over a single row. Figure 7 and Figure 8 show tabular processing

to design capacitances using poly-poly-2 CMOS 500 nm technology. Figure 5 starts with the value of the capacitance and ends up with squared plates in Poly-Poly to design the capacitance. Figure 7 starts with L/2 feature size (300 nm in this case) X-Y rectangular dimensions to obtain the effective capacitance value. Figures 9 and 10 show the CMOS 500 nm IC layouts of 2 pF and 10 pF capacitances, obtained from Electric-VLSI. The artwork was also developed once the Excel methodology was used to evaluate the conceptual design. Capacitors from Figures 8 and 9 are used for operational transconductance amplifier (OTA) compensation later in this paper.

-									
From C	p-p to X-Y	dimensio	ns						
Req	Cpoly2-poly C5-Mod Cpoly-sub			Estimated	Dimensi	ons 1x1	Dimens	Parasitic	
Ср-р	Poly2-Poly	λ	Poly-Sub	Area sq(μμ)	Х	Υ	X	Y	Cp-s
рF	aF/µm^2	μm	aF/µm^2	μm^2	λs	λs	λs	λs	fF
0.4	922	0.3	86	4820.44	69.4	69.4	138.9	34.7	37.31
0.5	922	0.3	86	6025.55	77.6	77.6	155.2	38.8	46.64
1	922	0.3	86	12051.10	109.8	109.8	219.6	54.9	93.28
1.5	922	0.3	86	18076.64	134.4	134.4	268.9	67.2	139.91
2	922	0.3	86	24102.19	155.2	155.2	310.5	77.6	186.55
3	922	0.3	86	36153.29	190.1	190.1	380.3	95.1	279.83
5	922	0.3	86	60255.48	245.5	245.5	490.9	122.7	466.38
7.5	922	0.3	86	90383.22	300.6	300.6	601.3	150.3	699.57
10	922	0.3	86	120510.97	347.1	347.1	694.3	173.6	932.75
	$\lambda$ = SCA	LE SIZE	OF THE	ELECTRIC	VLSI L	AYOU	r proc	GRAM	

## **Processing Flow**

Figure 7. Tabular method to design CMOS capacitances. This Excel goes from capacitance value to X-Y dimensions.

## 4.3. Two-Dimensional Processing Excel Method

In two-dimensional processing, the method proceeds with the horizontal first row having the specifications and CMOS technology parameters of the conceptual design. Each row will define a step in the processing design flow such that the Excel will progress down and away from the first cell of the spreadsheet. The evaluations are arranged such that intermediate calculations follow a slope down from the early decisions all the way to the last decision. Usually, an iterative process is necessary to comply with two or three specifications with a single degree of freedom. For instance, with the bias current,  $I_{SS}$ , of a differential amplifier, can fulfill expectations for Slew Rate (SR), Output Resistance  $(R_{out})$ , and Power dissipation  $(P_{diss})$  in the conceptual design of a CMOS differential amplifier. Figure 11 illustrates the Excel method to develop the conceptual design for a CMOS differential amplifier. The method illustrates the step-by-step evaluation and decision-making process downward, and the CMOS technology specifications are shown rightwards as shown in Figure 11. The CMOS Technology characteristics flow horizontally to the right and the CMOS design equations, from (5) to (10), flow downwards illustrating the step-by-by step procedure. The evaluation of CMOS transistor size, W/L, goes along to fulfill the required specification. However, if a particular spec does not convince the design engineer, the processing flow can be stopped, and a recalculation with a different spec or different decision making is readily possible at every row. The processing flow continues row/column by row/column until the final step and size selection is terminated.

From )	(-Y dime	ensions to	Ср-р								
Dimen	sions	Cpoly2-poly	C5-mod	C from C5	Estimated	Equivalent	Parasitic				
X	Y	Poly2-Poly	λ	Poly-Sub	Area sq(μμ)	Ср-р	Cp-s				
λs	λs	aF/µm^2	μm	aF/µm^2	μm^2	рF	fF				
68	68	922	0.3	<mark>86</mark>	416.16	0.38	35.79				
78	78	922	0.3	86	547.56	0.50	47.09				
136	136	922	0.3	86	1664.64	1.53	143.16				
157	157	922	0.3	86	2218.41	2.05	190.78				
670	180	922	0.3	86	10854.00	10.01	933.44				
200	200	922	0.3	86	3600.00	3.32	309.60				
500	120	922	0.3	86	5400.00	4.98	464.40				
68	68	922	0.3	<mark>86</mark>	416.16	0.38	35.79				
68	68	922	0.3	<mark>86</mark>	416.16	0.38	35.79				
170	170	922	0.3	<mark>86</mark>	2601.00	2.40	223.69				
112	108	922	0.3	86	1088.64	1.00	93.62				
.λ= S	$\lambda$ = SCALE SIZE OF THE ELECTRIC_VLSI LAYOUT PROGRAM										

# **Processing Flow**

Figure 8. Tabular method to design CMOS capacitances. This Excel sheet goes from X-Y dimensions to capacitance value.







Figure 10. CMOS IC poly-poly 10 pF capacitance with rectangular X-Y layout from Electric-VLSI.

	Differer	ntial Amp	plifier St	ep-by-Ste										
	CMOS 60	1OS 600nm ProcessUses MOSIS data from C5_Models latest runs												
	VT0	VDD	VSS	SR (V/μ)	Av (V/V)	CL (pF)	f_3dB	λ	minICMR	maxICMR	Pmax	Кр	Kn	
	0.8	2.5	-2.5	10	50	2	1E+05	0.04	-1	2	1	37.4	118	
ste	0	lss	max lss	Min Iss	Pssmax	Rout_max	lss-sel	Vsg3	(W/L)3	(W/L)1	Vds5	(W/L)5	Vgs5	Vbias_
1	Iss-for-SR	2.00E-05												
2	Max_lss		2.00E-01		1.00E-04									
3	Min_lss			3.14E-05		7.96E+05								
seleo	t Iss						3.50E-05							
4	Ic_MAX							1.30	3.74					
selec	t (W/L)3								4.00					
5	Gain									1.19				
sele	t (W/L)1									1.00				
6	Ic_MIN										0.15	4.98		
sele	t (W/L)5											5.00		
7	Vbias												0.95	-1.55

Process Specifications and Boundary Conditions

**Figure 11.** Two-dimensional processing method to design a differential amplifier. From the initial cell (top-left), the conceptual design progress downward, step by step, and to the right to size each transistor.

Figure 12 shows the schematic from Electric-VLSI of a differential amplifier using results from the Excel two-dimensional method. In this case, every value sizing the MOSFETs means L/2 times or half the feature size of the technology. For instance, the middle twin transistors have a size of W =  $120 \times (0.5/2) = 30 \mu m$  by L =  $2 \times (0.5/2) = 0.5 \mu m$ . Furthermore, the SPICE code to perform DC and AC testing in this conceptual design of the device is shown in Figure 12. Once this schematic circuit model is tested, the next step is developing the layout. Differential amplifiers are the core of every instrumentation amplifier and their layout must be considered very carefully. Figure 13 shows the strategy recommended by J. Baker [4] to develop a common centroid layout.

With careful development by considering the DRC (direct rule checking) from Electric-VLSI CMOS 500 nm kit, the layout shown in Figure 14 is developed by using the evaluations and conceptual model obtained from the Excel two-dimensional method.



Figure 12. A differential amplifier schematic using CMOS 500 nm technology.



Figure 13. Common centroid layout recommended for big matched differential pair transistors [3,4].



**Figure 14.** Conceptual design layout (using Electric-VLSI) of a differential amplifier from an Excel method.

## 5. Methodologies for Complete Amplifier Design

The Excel methodologies shown previously can be applied to develop conceptual designs of complete functional blocks such as a cascode amplifier with its bias circuit, and a two-stage operational transconductance amplifiers (OTA). The OTA amplifier includes a compensation capacitance which is necessary to ensure stability and reliable operation for the required frequency response. Even though the following design examples are not very specialized, the literature shows many examples of more specialized conceptual designs where microelectronic design has been extended [30–33]. The design flow that students

must follow to perform the conceptual design of the microelectronic device is given as follows:

- 1. Select or create the basic structure. This step consists of obtaining the schematic showing the transistors and their connections. This diagram does not change through the design process unless the specs cannot be met.
- 2. Set up the Excel spreadsheet to show the specifications and boundary conditions in the first line. Make sure that the units are consistent all the way through the design process. Develop the decision process line by line using the equations and specs. The design flow goes downwards.
- 3. Select the DC currents and transistor sizes to meet specs. This is where the major design effort will go. Simulators are used to aid the designer in this phase. However, a rough performance of the circuit should be known a priori.
- 4. Physical implementation. Layout of the transistors, floor-planning the connections, pinouts, power supply buses, and grounds. Extraction of physical parasitic and re-simulation. Verify that the layout is a physical representation of the circuit.
- 5. Furthermore, in designing a multi-stage amplifier where two or three stages integrate the complete device, the following considerations are made:
  - I. The characterization of the microelectronic device is the initial fundamental step in the analysis and design of the multi-stage amplifier.
  - II. Ideal analysis provides the first insight to the circuit operation by performing circuit analysis over the external components.
  - III. Practical models of the op-amp include static and dynamic parameters.
  - IV. The Excel method helps in developing the overall step-by-step procedure which accounts for all the requirements and specifications
  - V. The design of multi-stage devices involves the following boundary conditions and requirements:
    - a. Boundary conditions: CMOS technology, process specs ( $V_{T0}$ ,  $C_{ox}$ , K'), supply voltage and current range ( $V_{DD}$ ,  $V_{SS}$ ,  $I_{SS}$ ), operating temperature ( $T_o$ ) and range.
    - b. Requirements: Gain  $(A_v, A_i)$ , gain bandwidth (GB), settling time  $(T_s)$ , slew rate (SR), input common mode range (ICMR), common mode rejection ratio (CMRR), power supply rejection ratio (PSRR), output voltage swing  $(v_{out} (max), v_{out} (min))$ , output resistance  $(R_{out})$ , offset voltage  $(V_{OS})$ , noise  $(e_{out}^2)$ , layout area.
    - c. Verify that intermediate stages couple correctly without causing instabilities such as the influence of mirror poles in the transfer function. The Excel method can be used to revise and iterate the process in search of a robust device.
  - VI. Compensation techniques involve: Miller, feed-forward, and selfcompensation schemes.

## 5.1. Cascode Amplifier with Bias Source

The cascode amplifier obtains a higher gain and output resistance than the traditional inverting amplifier stages. Typical design parameters are slew rate (SR), output swing, and power dissipation for a simple cascode stage. Figure 15 illustrates the Excel method to develop the conceptual design for a cascode amplifier. As mentioned before, the method illustrates the step-by-step evaluation and decision-making process downwards and the processing flow goes rightwards. Again, the CMOS Technology characteristics and specifications flow horizontally to the right and the CMOS design equations, from (5) to (10), flow downwards illustrating the step-by-by step procedure. The evaluation of CMOS transistor size, W/L, goes along to fulfill the required specification.

	- 1							_		_		_	_		
р			Three Tr	ransistor (	Cascode A	mplifier									
~			CMOS 60	Onm Proces	sUses M	OSIS data fr	rom C5_Mor	dels latest r	uns						
\$			VT0	VDD	VSS	SR (V/μ)	Av (V/V)	CL (pF)	λ	Vo min	Vo max	Pmax	Кр	Kn	
i			0.8	5	0	10	50	5	0.03	1.5	4	1	37.8	115.6	
e.		step		I-DS	max-ID	min-ID	Pssmax	ID-sel	(W/L)3	VGG3	(W/L)1	Vds1	Vds2	(W/L)2	VGG2
n		1	ID for SR	5.00E-05											
		2	Max ID		2.00E-04		1.00E-03								
P		3	Min ID	5.00E-05		5.00E-05									
r		select	ID					5.00E-05							$\square$
٥		4	Vo Max						2.65	3.26					$\square$
c		select	(W/L)3						3.00	3.2					$\square$
e		5	Gain								0.49				$\square$
s		select	(W/L)1								1.00				$\square$
s		6	Vo Min									0.93	0.57	2.66	2.27
		select	(W/L)2											3.00	2.30

## Process Specifications and Boundary Conditions

Figure 15. Two-dimensional processing method to design a CMOS cascode amplifier.

Figure 16 shows the schematic from Electric-VLSI of the conceptual design of the cascode amplifier which includes the bias network on the left of the stacked three MOSFET from the right. The bias circuit was implemented and evaluated in a separate analysis using the guidelines from J. Baker [4]. Once this schematic circuit model is tested, the next step is to develop the layout.



Figure 16. A three stack cascode amplifier schematic using CMOS 500 nm technology.

The layout is developed using Electric-VLSI and includes both the bias circuit and the three stacked transistors as shown previously in Figure 16. Figure 17 illustrates the layout design with the corresponding  $V_{dd} = 5$  V and Vss = 0 (ground). This layout shows the transistors in horizontal layouts where the lower two levels include N-type transistors, and the upper level includes the P-type transistors which appear in the circuit schematic



from Figure 16. The layout also shows the p-well and n-well over the lower two N-channel transistor levels and the upper P-channel transistor level, respectively.

Figure 17. Conceptual design layout (using Electric-VLSI) of a three stage cascode from an Excel method.

### 5.2. OTA with Miller Compensation

The development of a conceptual model for an operational transconductance amplifier with Miller compensation has an additional complexity of calculating the feedback capacitance that operates the amplifier in a stable and reliable regime. The phase margin  $PM > 50^{\circ}$  and the sizing of more than 10 transistors make the Excel method larger. Figure 18 shows the Excel spreadsheet workout with the evaluation of transistor sizes and dominant pole calculations.

Again, the procedure shows the conceptual design strategy mentioned before, the CMOS technology characteristics and specifications flow horizontally to the right and the CMOS design equations, from (5) to (10), flows downwards illustrating the step-by-by step procedure. In this case, Figure 18 has two downward decision flows. The first one evaluates the design with the sole calculation of the Miller compensation capacitance to provide the required phase margin and stability criteria [3,4,17,27]. The second design decision flow determines the size of a transistor to locate a right-hand side pole (RHP) exactly to cancel the second pole. This way the dominant pole will be extremely alone well inside the gain bandwidth and the amplifier will have, even higher, phase margin PM. Figure 19 shows the Electric-VLSI layout schematic of the conceptual design obtained for the 500 nm CMOS technology. Figure 20 illustrates the layout of the OTA amplifier with the area dominance of the compensation capacitance  $C_c = 3$  pF. Those capacitances are extremely large and, in this case, the layout generated has a squared shape, like the ones developed with a single straight Excel methodology for capacitors. The layout shows also the common centroid differential amplifier stage developed before and the big, 125  $\mu$ m/1  $\mu$ m, p-channel MOSFET transistor right above the output voltage of the amplifier.

	Cox	VT0	VDD	VSS	SR (V/µ)	Av (V/V)	CL (pF)	GB	λp	λn	minICMR	maxICMR	Pmax	Кр	Kn	Vo min	Vo max	PM
	0.25	0.8	2.5	-2.5	10	3000	10	5.00E+06	0.03	0.03	-1.6	2	1	37.8	115.6	-2	2	60
1	step		Cc	15	(W/L)3	p3-rad	p3-Hz	10*GB	gm1	(W/L)1	Vds5	(W/L)5	gm6	gm4	(W/L)6	1-6	Pdiss	(W/L)7
	1	min Cc	2.20														mW	
s	select		3															
	2	15		3E-05														
s	elect			3E-05														
	3	maxICM	R		19.84													
\$	elect				20													
	4	check M	l p			2.25E+10	3.58E+09	5.00E+07										
c	:heck	p3>10G	P				p3-Hz>>	10*GB										
	5	gm1	(W/L1)						9.42E-05	2.56								
s	elect									3								
	6	Vds5	(W/L)5								-0.344	4.38						
s	elect											5						
	7	gm6	(PM)	(W/L)6									9.42E-04	1.51E-04	125.10			
\$	elect														125			
	8	16	Pdiss													9E-05	0.62	
cl	heck																ОК	
	9	(W/L)7																15.65
54	elect																	16
	10	check V	o min								Av DB			Va min		-2.18		
Γ		check A	v					Av	17500		84.861							
R	RHP-ZE	RO CON	PENSA	TION US	ING AN	ADDITION	AL TRANSI	STORS: M8	, M9, M10	AND M11								
	11	19,110,1	11		select cu	rrent	3.00E-05		3.00E-05		(W/L)11 =		39.94					
5	elect	(W/L)11											40					
	12	(W/L)9					(W/L)10=	39.94			(W/L)9 =		5					
s	elect							40					5					
F	13	(W/L)8									(W/L)8 =		28.87					
s	elect										1.1.1.1		30					
Ê	14	check		V5G10	1.00	RZ	4600.1415	71	-9.42E+07	-1.499E+07		02	-94200000		OK for P	HP-ZERO (	COMP	

### **Process Specifications and Boundary Conditions**

Figure 18. Conceptual OTA design with two downward decision flows.







Figure 20. Electric-VLSI layout of the CMOS OTA with RHP-zero compensation.

### 5.3. Performance Simulation Tests

The Excel methods are used to synthesize the conceptual designs of integrated circuits and devices to teach and develop successful strategies that can be repeated for different CMOS technologies. This section will compare the expected specifications defined initially with the schematic and layout results from SPICE simulations run by Electric-VLSI using the LTSpice program as a kernel.

Four conceptual design cases are analyzed which are part of a formal course in microelectronics [28]. The design problems are:

- 1. Differential amplifier using common centroid layout.
- 2. The three-stack MOSFET cascode amplifier using the corresponding bias power reference.
- 3. The two-stage OTA stage with Miller compensation and having the RHP cancel the second dominant pole.
- 4. A three-stage op-amp using shunt feedback output stage to enhance the output resistance.

The differential amplifier results are summarized in Table 5. This amplifier stage is shown in Figures 11 and 13 and resolved using the two-dimensional processing method from the Excel methodology illustrated in Figure 10. The results comply with all the specifications established for the conceptual model using the CMOS 500 nm technology as illustrated in Table 5. Some parameters differ slightly because of the iterative nature of the design decision flow in which the number of requirements is higher than the number of degrees of freedom available: three transistor sizes (W/L) and the operating Q point of the current-sinking at the lower transistor in Figure 13.

Table 5. Specifications and measured values from simulation tests in the CMOS Differential amplifier.

Specification	Requirement	Simulation Measured Value
CMOS process	0.5 μm	0.5 μm
Supply voltage	5 V rail to rail	$\pm 2.5 \text{ V}$
Supply current	>30 µA	49.2 μΑ
Gain	>30 dB	33.7 dB
Offset voltage	<20 mV	19.57 mV
f <sub>-3dB</sub>	$>100 \text{ KH} (C_L = 2 \text{ pF})$	812.8 KH
Slew Rate	$>10 \text{ V}/\mu \text{s}$ ( $C_L = 2 \text{ pF}$ )	$-25\mathrm{V}/\mathrm{\mu s}$ and $+27\mathrm{V}/\mathrm{\mu s}$
Power Dissipation	<1 mW	0.492 mW

Note: Differences are between accepted tolerances of +/-10% for the conceptual model development.

Table 6 illustrates the results for the cascode amplifier conceptual model developed using the Excel method shown in Figure 14. Again, this amplifier stage, shown in Figures 15 and 16, describes the fulfillment of all the specifications established for the conceptual model using the CMOS 500 nm technology. The power dissipation shown for this amplifier includes the three stack of transistors and the bias reference network shown in Figures 15 and 16. Again the number of requirements is higher than the number of degrees of freedom available: three transistor sizes (W/L) and the operating Q point of the current-sinking at the lower transistor in Figure 16. The two previous amplifiers, differential, and cascode are not used as independent amplifiers, but they are part of a larger multistage amplifier or microelectronics functional block. Therefore, the conceptual models for a highly specialized microelectronic device contain 2, 3, or more of those primitive amplifiers described previously. Now we will present results for the conceptual model of a two-stage OTA and of a three-stage operational amplifier. Table 7 shows results for the conceptual model of a two-stage OTA that includes a differential amplifier and an inverting amplifier that drives the load. This was the amplifier illustrated by Figures 18 and 19 and developed using the methodology of Figure 17. This OTA conceptual design includes additional specifications such as offset voltage, output swing, phase margin, power supply rejection ratio (PSRR), gain-bandwidth GB, and settling time. This is a preview of the project that the students in the microelectronics course develop at the end of the semester.

Specification	Requirement	Simulation Measured Value				
Supply voltage	5 V rail to rail	$V_{dd} = +5 \text{ V} \text{ and } V_{ss} = 0 \text{ V}$				
Supply current	>30 µA	52 µA				
Gain	>26 dB	31.1 dB				
f <sub>-3dB</sub>	$>200 \text{ KH} (C_L = 5 \text{ pF})$	361 KH				
Slew Rate	$>10 \text{ V}/\mu \text{s} (C_L = 5 \text{ pF})$	$-300 \text{ V}/\mu\text{s}$ and $+10.45 \text{ V}/\mu\text{s}$				
Output swing	from $1.5$ to $4.0$ V	from 0.66 to 4.03 V				
Power Dissipation	< 1  mW	0.996 mW				

**Table 6.** Specifications and measured values from simulation tests in the CMOS cascode amplifier using using  $0.5 \mu$  porocess.

Note: Differences are between accepted tolerances of +/-10% for the conceptual model development.

**Table 7.** Specifications and measured values from simulation tests in the CMOS 2-stage OTA using  $0.5 \mu$  process.

Specification	Requirement	Simulation Measured Value
Supply voltage	5 V rail to rail	±2.5 V
Supply current	>30 µA	30 µA
Gain	>80 dB	80 dB
Gain-Bandwidth	>10  MH	14.7 MH
Slew Rate	$>10 \text{ V}/\mu \text{s}$ ( $C_L$ = 2 pF)	$-10$ V/ $\mu$ s and + 12 V/ $\mu$ s
ICMR	$> \pm 1.5 ~ m V$	-2.3  to + 2.4  V
Offset voltage	$<\pm 2~\mathrm{mV}$	-0.164  mV
PSRR	>70 dB	80 dB
Output Swing	$>\pm 2$	-2.25 V and $+2.25$ V
Phase Margin	$>70^{\circ}$	83°
Power Dissipation	< 1  mW	0.896 mW

Note: Power dissipation includes the bias current for both stages.

Finally, Table 8 illustrates the results obtained from a conceptual design of a 3-stage op-amp using an additional third stage with a shunt feedback scheme to reduce the output resistance of the device. The results are good with a low value with the negative slew-rate (SR) of  $-9 \text{ V/}\mu\text{s}$  which needs to be improved from this conceptual design developed using the methodologies discussed in this paper.

**Table 8.** Specifications and measured values from simulation tests in the CMOS 3-stage op-amp using  $0.5 \mu$  process.

Specification	Requirement	Simulation Measured Value
Supply voltage	5 V rail to rail	±2.5 V
Supply current	$>30 \ \mu A$	73 µA
Gain	>90 dB	80.19 dB
Gain-Bandwidth	>10 MH	24.9 MH
Slew Rate	$>10 \text{ V}/\mu \text{s}$ ( $C_L = 2 \text{ pF}$ )	$-9V/\mu s$ and + 12.45 V/ $\mu s$
ICMR	$>\pm1.5~{ m V}$	-2.2 to $+1.84$ V
Offset voltage	$<\pm 2$ mV	0.036 mV
PSRR	>70 dB	80 dB
Output Swing	$>\pm 2$	-2.25 V and + 2.25 V
Phase Margin	$>70^{\circ}$	$126.4^{\circ}$
Power Dissipation	< 1  mW	1.095 mW

Note: Power dissipation includes the bias current for the three stages.

The schematic of the three-stage op-amp conceptual design is shown in Figure 21. This figure illustrates the main sub-components of the device: bias and reference voltage circuit, first stage differential amplifier, Miller and right-hand plane zero (RHP) compensation for maximum stability, inverting amplifier second stage, and the output push/pull with shunt feedback differential amplifier that provides a lower output resistance in the device.



Figure 21. Electric-VLSI schematic of the 3-stage CMOS op-amp with RHP-zero compensation.

The conceptual design developed in this paper shows one of the major steps in designing analog integrated circuits (IC) for electronic instrumentation devices required by electronics, biomedical, robotics, and computer engineering majors. In analog IC design, a good combination of function or application with IC technology is necessary to obtain a successful solution. The Excel methodologies are an additional tool to validate and verify the conceptual model required before the device is sent to the foundry facility. Analog IC design consists of three major steps [3]: electric design, physical design (layout), and test design (testing). Engineers and designers must be flexible, use techniques such as the Excel methods, and have a skill set that allows them to simplify and understand a complex conceptual design problem. In microelectronics, device IC design is driven by improving technologies rather than new technologies [3]. The engineer should be aware that sometimes analog systems applications, where speed, area, or power, have certain advantages over the digital approach. Even using Excel methods, deep-submicron (DSM) technologies offer great challenges to the creativity of engineers and designers of IC microelectronic devices.

### 6. Analysis of Results

To further increase the potential of using Excel methods in developing instrumentation amplifiers for biomedical applications, we examined a project case study using the methodology to design a 3-stage amplifier having a low voltage and low power operation in the strong inversion zone. Undergraduate students during the spring semester of 2021 in the microelectronics course at Tecnológico de Monterrey developed a project where they used the methods learned in class [34]. This design was going to be used as a subsystem of a bioinstrumentation amplifier required in sensor signal conditioning applications.

The project consisted of the development of a three-element instrumentation amplifier for biomedical applications. Figure 22 illustrates the basic scheme where two op-amps receive the differential mode input signals, and a third op-amp changes the signal from differential mode to single-ended mode referenced to ground.



Figure 22. Classic topology for the instrumentation amplifier used in biomedical applications.

In Figure 22, each op-amp (OA1, OA2, and OA3) must be selected from possible topologies seen in class to achieve certain performance characteristics. Those op-amps have the following components:

- 1. First stage differential amplifier;
- 2. Second stage common source amplifier;
- 3. Third stage push-pull output stage;
- 4. Power source to provide the required bias currents and voltages.

Figure 23 shows a block diagram of each op-amp with all the functional parts of the system. For the power source, the students have the option of developing a high-performance bandgap reference source that is stable with respect to variations in supply voltage, temperature, and noise.



Figure 23. Overall block diagram of each operational amplifier.

The requirements and specifications provided in Table 9 are to be met for each of the op-amp blocks.

Specification	Requirement	Comment
CMOS Process	0.5 μm	
Supply Voltage	From 2.0 V to 5.0 V rail to rail	+/-1.0 V to $+/-2.5$ V (for Vdd/Vss)
Supply Current	<100 µA	
Temperature range	0 to 70 $^{\circ}$ C	
Gain	>90 dB	
Gain Bandwidth	>10 MH	
Settling time	<0.5 µs	
Slew rate	>10 V/ µs	Average of SR+ and SR-
ICMR	>0.8 V to $>+/-1.5$ V	Depends upon the supply rails
PSRR	>70 dB	
Output swing	>+/-2 V	
Output resistance	<100 Ohms	
Offset Voltage	From $+/-0.8$ V to $+/-2$ mV	Depends upon the supply rails
Noise	<50 nV/sqrtH at 100 H	
PM	>70°	
Power dissipation	<1 mW	

**Table 9.** Proposed specifications for the op-amp design for biomedical applications.

The students began with a literature survey before selecting the device to develop from the conceptual requirements, theoretical development, schematic development, and layout implementation using Electric\_VLSI [35-38]. Instead of adding a third stage at the output of the two-stage OTA device, they added a second stage differential-amplifier with compensation between the first differential stage and the output stage. This results in improving the frequency response and sacrificing some gain and having a better phase margin for even robust design. Even with this modification, the simulations show an increase in the overall gain compared to the last three stage designs (Section 5). The compensation capacitor *Cc* and *Rz* are kept for both differential stages, and the output terminals of both compensation elements are connected all the way to the output of the circuit. The methodology generates the transistor dimensions and capacitor values illustrated in Tables 10 and 11.

Table 10. Transistor dimensions for the new three-stage conceptual design.

Transistor Dimensions (W/L) Considering L = 500 nm					
M1, M2	8				
M3, M4	7				
M5	5				
M6	130				
M7	22.5				
M8	27				
M9, M12	5				
M10, M11	15				

Table 11. Capacitors and resistors for the new three-stage conceptual design.

Cs are in pF and Rs are in KW					
CL	2				
$C_{C}$	0.5				
R <sub>Bias</sub>	265				
$R_z$ (Implemented by M8)	5.22				

Considering the transistor dimensions for this new three stage amplifier design, the first cut expected parameters are calculated as follows, assuming long channel models.

1. Slew rate.

$$SR = \frac{I_5}{C_c} = \frac{10\mu}{0.5p} = 20 \text{ V/}\mu\text{s}$$
(17)

2. Estimated amplifier gain.

$$A_{v0} = \left(\frac{g_{m2}}{g_{m2} + g_{m4}}\right)^2 \cdot \frac{g_{m6}}{g_{ds6} + g_{ds7}} = \left(\frac{g_{m2}}{I_5/2(\lambda_2 + \lambda_4)}\right)^2 \cdot \frac{g_{m6}}{I_6(\lambda_6 + \lambda_7)}$$
(18)

$$g_{m2} = \sqrt{2\beta_2 I_2} = \sqrt{2 \times 115.6\mu \times 8 \times 5\mu} = 96.17 \ \mu S$$
 (19)

$$g_{m6} = 942.5 \ \mu S$$
 (20)

$$A_{v0} == \left(\frac{96.16\mu}{5\mu(0.04+0.05)}\right)^2 \cdot \frac{942.5\mu}{92.14\mu(0.04+0.05)} = 134.3 \text{ dB}$$
(21)

3. Power dissipation.

$$P_{diss} = (2I_5 + I_7)(V_{dd} + |V_{ss}|) = (2 \times 10\mu + 92.14\mu)(2 \times 1.8) = 403.7 \,\mu\text{W}$$
(22)

4. Output swing.

$$V_{out}(min) = V_{ss} - V_{DS7}(sat) = -1.8 + \sqrt{\frac{2I_7}{k_N(W/L)_7}} = 1.8 + \sqrt{\frac{2 \times 92.14\mu}{115.6\mu \times 22.5}} = -1.53 \text{ V}$$
(23)

$$V_{out}(max) = V_{dd} - V_{SD6}(sat) = 1.8 - \sqrt{\frac{2I_6}{k_P(W/L)_6}} = \sqrt{\frac{2 \times 92.14\mu}{37.8\mu \times 130}} = 1.607$$
(24)

5. Input common mode range, ICMR.

$$(W/L)_5 = \frac{2I_5}{k_N V_{DS5}(sat)^2} = \frac{2 \times 10\mu}{k_N \left(-V_{ss} + ICMR^- - \sqrt{\frac{I_5}{\beta_1}} - V_{T1}(max)\right)^2}$$
(25)

$$ICMR^{-} = -0.93V \tag{26}$$

$$(W/L)_{3} = \frac{I_{5}}{k_{P}(V_{dd} + ICMR^{+} - |V_{T3}|(max) + V_{T1}(min))^{2}}$$
(27)

$$ICMR^+ = 1.69V \tag{28}$$

6. Output resistance.

$$R_{out} = \frac{1}{g_{ds6} + g_{ds7}} = \frac{1}{I_6(\lambda_N + \lambda_P)} = \frac{1}{92.14\mu(0.04 + 0.05)} = 120.5 \text{ K}\Omega$$
(29)

Afterward, the circuit schematic and layout are developed in Electric VLSI to perform simulations tests. The compensation capacitor is designed as a poly-poly2 capacitor. The area for the capacitor is 78 by 78 lambda, with a 47 fF parasitic capacitance. Figure 24 shows the Excel first cut approximation used for the first part of the conceptual design.

DESIGN	OF A LO	OW VOL	TAGE TWO	O STAGE O	TA USING I	=0.6 MICR	OMETER. T	HE FIRST	TWO RO	WS ARE S	PECIFICATI	IONS					
Cox	VTO	VDD	VSS	SR (V/μ)	Av (V/V)	CL (pF)	GB	λр	λn	minICMR	maxICMR	Pmax	Кр	Kn	Vo min	Vo max	PM
0.25	0.8	1.8	-1.8	20	3000	2	2.90E+07	0.05	0.04	-0.7	1.7	1	37.8	115.6	-1.6	1.7	60
step		Cc	15	(W/L)3	p3-rad	p3-Hz	10*G8	gm1	(W/L)1	Vds5	(W/L)5	gm6	gm4	(W/L)6	1-6	Pdiss	(W/L)7
1	min-Cc	0.44		(W/L)4												mW	
select		0.5															
2	15		0.00001														
select			0.00001														
3	maxICM	IR		6.613757													
select				7													
4	check N	1-p			2.19E+10	3.49E+09	2.90E+08										
check	p3>10G	Р				p3-Hz>>	10*GB										
5	gm1	(W/L1)						9.11E-05	7.17								
select									8								
6	Vds5	(W/L)5								0.19601	4.50						
select											5						
7	gm6	(PM)	(W/L)6									9.11E-04	5.14E-05	123.92			
select														130			
8	16	Pdiss													8E-05	0.3397	
check																OK	
9	(W/L)7																37.992
select																	35
10	check V	o min								Av DB			Vo min		-1.60		
	check A	v					Av	24267		87.7002							
RHP-ZE	RO CON	IPENSA	TION USIN	G AN ADD	ITIONAL TR	RANSISTOR	S: M8, M9,	M10 AND	0 M11								
11	19,110,	/11		select cur	rent	1.00E-05		1.00E-05		(W/L)11 =	:	15.41					
select	(W/L)11	L										15					
12	(W/L)9					(W/L)10=	15.41			(W/L)9 =		5					
select							15					5					
13	(W/L)8									(W/L)8 =		25.65					
select												27					
14	check		VSG10	0.99	RZ	5490.885	z1	-4.55E+08	-7.246E+07		p2	-5E+08		OK for	RHP ZER	O COMP	

**Figure 24.** Excel methodology for the first part two-stage OTA conceptual design, before connecting the third stage to achieve higher gain and robust conceptual design.

The Excel design spreadsheet shown in Figure 24 replicates the strategy which has been proposed in this article. The CMOS technology characteristics and specifications flow horizontally to the right and the CMOS design equations, from (5) to (10), Tables 1 and 2, flow downwards illustrating the step-by-by step procedure. Figures 25 and 26 illustrate the schematic and layout diagrams for the three-stage amplifier designed using the methodology shown and initiating with the conceptual design equations coming from the long channel model.



Figure 25. Schematic diagram from Electric\_VLSI for the three-stage low-power high-gain operational amplifier.



Figure 26. Layout diagram from Electric\_VLSI for the three-stage low-power high-gain operational amplifier.

The output simulation runs show results that comply with the major specifications required by the design. Figure 27 illustrates the layout frequency sweep where the dB gain, gain bandwidth, and the phase margin are displayed. The DC gain obtained is 91.7 dB, the gain bandwidth GB is 46 MH and the phase margin is now close to 94°, which makes a very robust device.



**Figure 27.** Frequency sweep using. AC analysis in SPICE for the three-stage low-power high-gain op-amp Layout.

Figure 28 shows the time domain simulation to verify the slew rate response. The graph shows  $SR^+ = 21.47 \text{ V/}\mu\text{s}$  and  $SR^- = -19.35 \text{ V/}\mu\text{s}$  which is much more than the required response.



**Figure 28.** Time domain response using transient analysis in SPICE for the three-stage low-power high-gain op-amp layout.

To summarize the complete results for all the theoretical and expected parameters, Table 12 compares:

- 1. Desired requirements for the conceptual design;
- 2. Theoretical calculations using the Excel method;
- 3. Simulation results of the schematic circuit;
- 4. Simulation results of the layout circuit.

**Table 12.** Comparison of specification, theoretical, schematic and layout results for the three-stage, low-power high-gain operational amplifier conceptual design.

Specification	Requirement	Theory Calculation	Schematic Simulation	Layout Simulation
I <sub>SS</sub>	10 µA	10 µA	9.72 μA	9.72 μA
$A_{v0}$	>90 dB	134.3 dB	91.84 dB	91.72 dB
GB	>30 MH	40 M	50.87 M	46.1 M
$SR_+$	$>20 \text{ V}/\mu$	*	21.7 V/μ	21.1 V/μ
$SR_{-}$	$> 20 \text{ V}/\mu $	*	19.6 V/μ	19.4 V/μ
PM	$> 70^{\circ}$	*	$109.5^{\circ}$	93.9°
$P_{diss}$	< 1  mW	0.403 mW	0.675 mW	0.661 mW
ICMR-	-0.8  V	-0.93 V	-0.816 V	-0.81  V
ICMR+	1.4 V	1.69 V	1.632 V	1.62 V
$V_{out}(max)$	1.4 V	1.61 V	1.35 V	1.34 V
$V_{out}(min)$	-1.4  V	-1.53  V	-1.35 V	-1.34  V
PSRR-	>70 dB	*	88 dB	88.1 dB
PSRR+	>70 dB	*	97 dB	97.9 dB
CMRR	>90 dB	*	128.8 dB	123.4 dB
Noise	$<0 \mathrm{nV}/\sqrt{H}$	*	$20.9 \text{ nV}/\sqrt{H}$	$20.9 \text{ nV}/\sqrt{H}$

(\*) no anticipated calculations were made for those parameters.

The results compare well to the specified requirements except for the maximum positive swing of the output signal which does not reach the specified value of 1.4 V. Furthermore, the power dissipation is slightly higher than anticipated by the theoretical calculations of 0.403 mW. However, very good results are obtained in gain bandwidth (GB), CMRR, PSRR, phase margin PM, and noise. The requirements for slew rate (SR) and DC gain ( $A_{v0}$ ) are barely achieved. Some of those specifications could have been obtained with additional calibration and refinements in the final conceptual design.

The student team [34] decided to compare the conceptual design of this low-power operational amplifier with one that appears in the literature with similar characteristics and is used in similar biomedical applications [35]. Table 13 shows this comparison where the supply voltage and the power consumption appear higher in our design. The load capacitance was a specification given by the instructor. Slew rate and bandwidth show a significantly higher performance in our design; however, those were parameters also required by the instructor. The chip area is twice in our design, however, in this case, the instructor did not have any restrictions here and no optimization was performed to reduce the layout artwork whatsoever. Table 13 results, however, illustrate comparisons under different specifications such as the load capacitance and voltage supply. Reference [35] used  $C_L$  = 30 pF and ±1.65 V, whereas our example used  $C_L$  = 2 pF and ±1.8 V, respectively. Some of those conditions were imposed by the course instructor. These changes produced large differences in closed-loop bandwidth and slew rate (SR) as seen by Table 11 results. Furthermore, the power dissipation and chip area obtained here are twice the values obtained by reference [35]. Another way to reduce the power dissipation and voltage supply could have been if they have designed the amplifier to operate in the subthreshold zone. However, the model equations in the Excel method would need to be changed to include the subthreshold model whatsoever. This comparison is an excellent way to encourage confidence in students about the design of integrated circuits at the nanoscale level [36].

**Table 13.** Comparison of the three-stage op-amp conceptual design with Lópes-Martin´s design [35] which appears in the technical literature using CMOS 500 nm technology.

Parameter	Lópes-Martin et al. [35]	This Study
CMOS Technology µm	0.5	0.5
Supply V	$\pm 1.65$	$\pm 1.8$
Power consumption mW	0.26	0.67
Load capacitance pF	30	2
SR V/µs	2	20.19
Silicon area mm <sup>2</sup>	0.02	0.04
Gain bandwidth MH	1	46.06

A system simulation test was performed by students with an instrumentation amplifier working with a gain of 60 dB or 1000 v/v using the traditional three op-amp topology. Figure 29 shows the Icon-View of the simulation experiment having three op-amps (3S\_OA) and the required resistors to provide the gain factor and the bias of the device. The figure shows on the left-hand side the SPICE code to run the frequency domain test (.ac dec 100 0.1 1 G). Furthermore, the MOSFET models command < include C5\_models.txt> describes the 500 nm technology used in this case.

The frequency-domain tests show a DC gain very close to the required for this instrumentation amplifier. Figure 30 shows the output voltage graph with a measured gain of 59.7 dB, approximately 966 V/V, which shows less than 5 % error with respect to the required 1000 V/V gain. This gain goes along to up to 10 KH in bandwidth.

Finally, a time-domain system test for the instrumentation amplifier was developed to find the maximum output symmetrical swing for the device when amplifying the biomedical sensor signal. This transient test was performed with a 1.8 mV amplitude at 1000 H differential mode signal (V2-V1) at the input. Figure 31 shows the output signal with a+1.7323 V to -1.735 V swing which also shows the gain of 966 *v*/*v* for the instrumentation amplifier configuration.



Figure 29. Instrumentation amplifier system simulation test using icon-view with Electric\_VLSI.



Figure 30. SPICE AC frequency domain test for the instrumentation amplifier.



**Figure 31.** Time domain test for the instrumentation amplifier, where the maximum rail to rail swing is shown.

The designed operational amplifier and the instrumentational amplifier have good features for biomedical portable sensor conditioning applications. ECG and EEG biopotentials can be processed, and a good front device can be implemented using the conceptual design generated using these methods. The design of micro-power-operational amplifiers to fulfill biomedical instrumentation characteristics has become a huge yet complicated research task with great opportunity areas for improvement and innovation. The design proposed by the students offers further improvement opportunities (like reducing output resistance) yet possesses interesting features that would hopefully serve for its application in sensor signal conditioning in biomedical instrumentation.

Another application of the low power operational amplifier is to perform signal conditioning in signals coming from CMOS-MEMS sensors [39–42] where the CMOS amplifiers are used as the standard electronic system maneuvering platform. For instance, in [42] the low power operational amplifier is used as a high input impedance instrumentation amplifier to condition signals coming from a micro-hotplate in either Pirani, Temperature, or Gas CMOS-compatible MEMS sensors. Three operational amplifiers, designed in this paper, are set up as instrumentation amplifiers like the one shown in Figure 29. This configuration is ideal for this multifunctional sensor platform application, particularly for the results obtained by the temperature sensor as shown in Figure 32. Figure 33 shows the new instrumentation amplifier configuration used to condition the signal coming from the multiplatform temperature sensor [42] to obtain the results shown in Table 14. Figure 34 illustrates the linear conditioning performed by the instrumentation amplifier designed using the low power OTA developed by the students.



Figure 32. Output voltage for a CMOS MEMS temperature sensor with suspended gap from reference [42].

Table 14. CMOS-MEMS si	gnal conditionin	g for tem	perature sensor.
------------------------	------------------	-----------	------------------

Temperature (°C)	V2 (V)	V1(V)	V2-V1 (V)	Gain (V/V)	Vref (V)	Vo (V)
-20	0.338	0.34	-0.002	100	0	-0.2
0	0.34	0.34	0	100	0	0
20	0.342	0.34	0.002	100	0	0.2
40	0.344	0.34	0.004	100	0	0.4
60	0.346	0.34	0.006	100	0	0.6



**Figure 33.** Instrumentation amplifier to condition the output from the temperature sensor described by the multiplatform of reference [42].



**Figure 34.** The linear signal conditioning performed by the instrumentation amplifier on signals coming from the CMOS-MEMS Temp Sensor Multiplatform [42].

The results obtained in this CMOS-MEMS sensor application confirm the possibility of using the EXCEL methods to perform first cut front end device design and testing. In

teaching microelectronics, sometimes the use of high-performance software tools deviates attention toward the final objective and competence development for electronics engineers. The analysis of results from the previous two examples shows that the use of EXCEL methods provides a fertile background to start conceptual design processes without requiring complex or expensive software which is not readily available in many universities around the globe. Finally, to account for parasitic capacitances and temperature effects over the operating point of the amplifiers, Appendix A provides additional equations that can be added to the Excel method to find those fluctuations when using Spice simulations of the device. Appendix A.1 shows the parasitic capacitance modeling and Appendix A.2 illustrates the temperature model of this technology.

### 7. Conclusions

This paper described methods for developing the conceptual design of microelectronic circuits before performing schematic and layout simulations. The Excel methods developed here are useful for following guidelines and to make design decisions during iterative processes of designing where the number of specification requirements is larger than the degrees of freedom available to maneuver the design. The "thinking model" for paper and pencil calculations is described with the major roles and variants in the computation of electrical variables and sizes of the transistors that integrate analog microelectronic devices.

The Excel method may include straight single dimension, tabular, and two-dimensional methods that allow the design engineer to go step by step in the decision flow which navigates downwards in the Excel spreadsheet. Several examples of conceptual designs are shown with the corresponding spreadsheet diagram, circuit schematic, and layout design to perform simulation tests. The examples include passive components: resistors and capacitors, functional subcircuits such as primitive amplifiers, complete specialized amplifiers such as OTA devices, and term projects where students research specialized devices for biomedical instrumentation applications. The full-blown methodology includes the Excel method, schematic development, layout implementation, and simulation test, and it is used for conceptual design development in microelectronics courses at undergraduate and graduate levels. The undergraduate courses include industrial partner participation to develop instrumentation systems specified by industrial needs [43]; as a result, these educational collaborations with other entities have been applied in our academic educational practices and/or by applying our new educational model TEC21 at the undergraduate level [44–47]. As mentioned earlier, the main contributions of this research are in teaching CMOS microelectronics in which the method helps the instruction of the following points:

- (1) Conceptual design evaluations using the traditional equations and preparation of layout implementation by setting up the schematic of the design.
- (2) Providing possible design solutions, with layouts developed and comparing results with the thinking model specs to validate the third phase of the microelectronics design process.
- (3) The methodology to develop complicated CMOS analog integrated circuits (IC) conceptual designs, specifically for undergraduate and graduate students.
- (4) Development of complex conceptual designs using simple, non-expensive, and readily available software.
- (5) Comparing a designed device with others published in the technical literature and used in biomedical instrumentation systems, particularly how far students are from major fundamental specifications and requirements such as gain, swing, frequency response, noise, and low power characteristics.
- (6) Applying the designed amplifiers to CMOS-MEMS experimental devices in linear signal conditioning of multiplatform such as compound structures including Pirani, temperature, and gas sensors [42].

Finally, with the approach to teaching CMOS microelectronics and the use of readily available software such as Excel, it is possible to align teaching to reach the search for open innovation.

Author Contributions: G.D.-A., Conceptualization, methodology, simulation examples, validation, simulation validation, literature review, writing—original draft preparation, writing—review and editing, data processing, quantitative data collection, resources; J.M.R.-D., review conceptualization, methodology; O.I.G.P., Literature search, validation of the experimental methodology, review of data validation and simulations, analysis, writing, and editing. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Acknowledgments: We wish to thank Steven Rubin and the crew that developed Electric-VLSI for the availability to have many professors, instructors and students working with this software in making conceptual designs of Microelectronic devices. Also, we want to acknowledge Analog Devices for the availability of its premier LTSpice software to professors, students and practicing engineers. Both software packages were very important in the validation of the conceptual designs developed by the Excel methods. Thanks must go to J. R. Baker and Philip Allen for their advice and workshops taken from 2005 until recently. Furthermore, we want to thank the Tecnológico de Monterrey, particularly the School of Engineering and Sciences (EIC), the Mechatronics and Electrical Engineering Department and the Computer Science Department for their faculty and student participation in the Microelectronic course TE-3034. In addition, a special thanks to the students who have participated in both courses, undergraduate and graduate level, since 2005 at Tecnológico de Monterrey, specially to Humberto González and Andrea Rodríguez who shared their term project design which was presented in this paper as a good example of using Excel methods to design microelectronic devices. Their participation enhanced the ideas and teaching strategies followed through the years while improving and refining the materials, simulation experiments and the fabless projects developed. Finally, the authors would like to acknowledge the Writing Lab, Institute for the Future of Education, Tecnologico de Monterrey, Mexico, for the financial and logistical support in the open access of this publication. This study belongs to the initiative R4C-IRG Reasoning for Complexity Interdisciplinary Research Group https://tec.mx/en/r4c-irg, accessed on 3 October 2021.

Conflicts of Interest: The authors declare no conflict of interest.

## Appendix A

### Appendix A.1. Parasitic Capacitances Modeling

In the analysis of parasitics inside the CMOS transistor, we can identify three types of capacitances. First, the thin oxide capacitance between the gate and the other terminals of the transistor that depends on the operating zone with an approximate value of

$$C_g = WLC_{ox} \tag{A1}$$

In the accumulation zone and factionary values of *Cg* of depletion, saturation, and linear zones of operation. The second type of parasitic capacitance inside the CMOS transistor is the PN junctions capacitors formed between the material P and N in the channel and substrate and sidewall capacitances. This is given by,

$$C_{j} \cong \frac{C_{jb}(A_{b} + A_{sw})}{\left(1 - \frac{v_{j}}{\varphi_{B}}\right)}$$
(A2)

The third type of parasitic capacitance inside a CMOS transistor is the Overlap capacitance that can be from the overlap of gates due to lateral diffusion (Cov) and fringing effects (Cf).

$$C_{ov} = C_{ox} \times L_D \tag{A3}$$

$$C_f = \frac{2\varepsilon_{ox}}{\pi} \ln\left(1 + \frac{T_{poly}}{t_{ox}}\right) \tag{A4}$$

$$C_{ol} = C_{ov} + C_f \tag{A5}$$

Table A1 illustrates the summary of the parasitic capacitances in a MOSFET at different operating zones. Moreover, Figure A1 depicts graphically the parasitic capacitance model of the CMOS devices.

fable A1. Parasitic MOSFET	capacitance at different o	peration zones.
----------------------------	----------------------------	-----------------

C/Zone	Cutoff	Linear	Saturation
Cgs	C <sub>ol</sub>	$C_{ol} + 0.5C_g$	$C_{ol} + 0.67C_{g}$
$C_{gd}$	Col	$C_{ol} + 0.5C_{g}$	C <sub>ol</sub>
$C_{gb}$	$[C_g C_{jc}]/[C_g + C_{jc}]$	0	0
C <sub>sb</sub>	C <sub>jsb</sub>	C <sub>jsb</sub>	C <sub>jsb</sub>
C <sub>db</sub>	C <sub>jdb</sub>	C <sub>jdb</sub>	C <sub>jdb</sub>
Subscripts: sb = source bulk	gb = gate bulk	db = drain bulk	ol = overlapp



Figure A1. The parasitic capacitance model of the CMOS transistors.

### Appendix A.2. Temperature Model and the Influence of Temperature in the Conceptual Design

In CMOS technology the influence of temperature in device conceptual designs is reflected mainly in the transconductance parameter and the threshold voltage, which can be evaluated using the Excel methods and validated using the ".temp" command in SPICE. The transconductance parameter changes with temperature as follows:

$$k(T) = k(T_0) [T/T_0]^{-1.5}$$
(A6)

The threshold voltage changes with temperature as follows:

$$V_T(T) \cong V_T(T_0) + \propto (T - T_0) \tag{A7}$$

Typically for NMOS transistors,  $\propto_{NMOS}$  varies from  $-2 \text{ mV/}^{\circ}\text{C}$  to  $-3 \text{ mV/}^{\circ}\text{C}$  from 200 °K to 400 °K and for PMOS transistors the sign is reversed as one may expect. Therefore, the overall evaluations and calculations performed using the Excel method at room temperature would need to be re-calculated to account for the temperature range of operation. This can be validated using SPICE simulation as mentioned above.

For example, assume that we want to determine how the drain-source operating current of a CMOS 0.500  $\mu$ m NMOS transistor varies from 27 °C to 100 °C. The device has  $W/L = 5 \mu/1 \mu$ ,  $k(T_0) = 117 \mu$ A/V<sup>2</sup>,  $V_{GS} = 2$  V,  $V_T(T_0) = 0.8$  and  $T_0 = 27$  °C. The following calculations are performed:

(a) At room temperature:

$$M_{DS}(27 \ ^{\circ}\text{C}) = \frac{117 \mu\text{A}/\text{V}^2 \cdot 5\mu}{2 \cdot 1\mu} (2 - 0.8)^2 = 421.2 \ \mu\text{A}^2$$

(b) At 100 °C (373 °K):

$$k(100 \ ^{\circ}\text{C}) = 117\mu\text{A}/V^{2}(373/300)^{-1.5} = 84.39 \ \mu\text{A}/V^{2}$$
$$V_{T}(100 \ ^{\circ}\text{C}) = 0.8 - 0.002(73^{\circ}\text{C}) = 0.654 \ \text{V}$$
$$I_{DS}(100 \ ^{\circ}\text{C}) = \frac{84.39\mu\text{A}/V^{2} \ \cdot 5\mu}{2 \cdot 1\mu}(2 - 0.654)^{2} = 382.23 \ \mu\text{A}$$

The previous example illustrates that a change of +73 °C in temperature produces a drop of 9.3% in the operating drain-source current of this device. Those calculations can be developed in the Excel methods to find the influence of temperature in the amplifier design. Those changes can also be verified by SPICE simulation using the temperature sweep accordingly.

## References

- 1. Rupp, K. 40 Years of Microprocessor Trend Data. Technische Universität Wien, Vienna Austria. Available online: https://www.karlrupp.net/2015/06/40-years (accessed on 10 May 2021).
- IBM Unveils World's First 2 Nanometer Chip Technology, Opening a New Frontier for Semiconductors. New York, NY. Available online: https://newsroom.ibm.com/2021-05-06-IBM-Unveils-Worlds-First-2-Nanometer-Chip-Technology,-Opening-a-New-Frontier-for-Semiconductors (accessed on 10 May 2021).
- 3. Allen, P.; Holberg, J. CMOS Analog Circuit Design, 3rd ed. Oxford University Press: Oxford, UK, 2011.
- 4. Baker, R.J. CMOS Circuit Design, Layout and Simulation, 3rd ed.; John Wiley & Sons: Hoboken, NJ, USA, 2019.
- 5. Verbeeck, J.; Van Uffelen, M.; Steyaert, M.; Leroux, P. Conceptual Design of a MGy tolerant Integrated Signal Conditioning in 130 nm and 700 nm CMOS. J. Instrum. 2012, 7, C01017. [CrossRef]
- 6. Henderson, S.C.; Johnson, E.W.; Janulis, J.R.; Tougaw, P.D. Incorporating standard CMOS design Process methodologies into the QCA logic design process. *IEEE Trans. Nanotechnol.* **2004**, *3*, 2–9. [CrossRef]
- 7. Lee, T. The Design of CMOS Radio Frequency Integrated Circuits, 2nd ed.; Cambridge University Press: Cambridge, UK, 2004.
- 8. Canovas-Carrasco, S.; Garcia-Sanchez, A.-J.; Garcia-Sanchez, F.; Garcia-Haro, J. Conceptual Design of a Nano-Networking Device. *Sensors* 2016, 16, 2104. [CrossRef] [PubMed]
- 9. Grasso, A.D.; Marano, D.; Palumbo, G.; Pennisi, S. Design Methodology of Subthreshold Three-Stage CMOS OTAs Suitable for Ultra-Low-Power Low-Area and High Driving Capability. *IEEE Trans. Circ. Syst. I Regul. Pap.* **2015**, *62*, 1453–1462. [CrossRef]
- Aghighi, A.; Atkinson, J.; Bybee, N.; Anderson, S.; Crane, M.; Bailey, A.; Morell, R.; Hassanin, A.; Tajalli, A. CMOS Amplifier Design Based on Extended g<sub>m</sub>/I<sub>D</sub> Methodology. In Proceedings of the 2019 17th IEEE International New Circuits and Systems Conference (NEWCAS), Munich, Germany, 23–26 June 2019; pp. 1–4. [CrossRef]
- 11. Dongyan, H.; Ming, Z.; Wei, Z. Design methodology of CMOS low power. In Proceedings of the 2005 IEEE International Conference on Industrial Technology, Hong Kong, China, 14–17 December 2005; pp. 114–118. [CrossRef]
- Green, M.M.; Pisani, M.B.; Dehollain, C. Design methodology for CMOS distributed amplifiers. In Proceedings of the 2008 IEEE International Symposium on Circuits and Systems, Seattle, WA, USA, 18–21 May 2008; pp. 728–731. [CrossRef]
- Hernández Sanabria, E.; Amaya Palacio, J.; Herrera, H.H.; Van Noije, W. A design methodology for an integrated CMOS instrumentation amplifier for bioespectroscopy applications. In Proceedings of the 2017 CHILEAN Conference on Electrical, Electronics Engineering, Information and Communication Technologies (CHILECON), Pucón, Chile, 29–31 October 2017; pp. 1–7. [CrossRef]
- Mohammed, M.; Abugharbieh, K.; Abdelfattah, M.; Kawar, S. Design methodology for MOSFET-based voltage reference circuits implemented in 28 nm CMOS technology. *AEU—Int. J. Electron. Commun.* 2016, 70, 568–577. [CrossRef]
- 15. ON-Semiconductors. *MOSIS Wafer Electrical Tests Run V37P, Technology SCN05, Feature Size 0.5 Microns*; Technical Report; University of California's Information Sciences Institute (ISI), University of Southern California: Marina del Rey, CA, December 2016.
- 16. Lewyn, L.L.; Ytterdal, T.; Wulff, C.; Martin, K. Analog Circuit Design in Nanoscale CMOS Technologies. *Proc. IEEE* 2009, 97, 1687–1714. [CrossRef]
- 17. Sadeqi, A.; Rahmani, J.; Habibifar, S.; Ammar Khan, M.; Mudassir Munir, H. Design Method for Two-Stage CMOS Operational Amplifier Applying Load/Miller Capacitor Compensation. 2020, pp. 153–162. Available online: https://mpra.ub.uni-muenchen. de/id/eprint/102931 (accessed on 1 September 2021).

- Cheah, S.L.Y.; Yuen-Ping, H.O.; Shiyu, L.I. How the effect of opportunity discovery on innovation outcome differs between DIY laboratories and public research institutes: The role of industry turbulence and knowledge generation in the case of Singapore. *Technol. Forecast. Soc. Chang.* 2020, 160, 120250. [CrossRef]
- 19. Pearce, J.M. Economic savings for scientific free and open source technology: A review. *HardwareX* **2020**, *8*, e00139. [CrossRef] [PubMed]
- Vicente-Saez, R.; Gustafsson, R.; Van den Brande, L. The dawn of an open exploration era: Emergent principles and practices of open science and innovation of university research teams in a digital world. *Technol. Forecast. Soc. Chang.* 2020, 156, 120037. [CrossRef]
- 21. Romero-Rodríguez, J.-M.; Ramírez-Montoya, M.-S.; Aznar-Díaz, I.; Hinojo-Lucena, F.-J. Social appropriation of knowledge as a key factor for local development and open innovation: A systematic review. *J. Open Innov.: Technol. Market Complex.* **2020**, *6*, 44. [CrossRef]
- Beck, S.; Bergenholtz, C.; Bogers, M.; Brasseur, T.-M.; Conradsen, M.L.; Di Marco, D.; Distel, A.P.; Dobusch, L.; Dörler, D.; Effert, A.; et al. The Open Innovation in Science research field: A collaborative conceptualisation approach. *Ind. Innov.* 2020, 1–50. [CrossRef]
- 23. Roman, M.; Liu, J.; Nyberg, T. Advancing the open science movement through sustainable business model development. *Ind. High. Educ.* **2018**, *32*, 226–234. [CrossRef]
- Vicente-Saez, R.; Martinez-Fuentes, C. Open Science now: A systematic literature review for an integrated definition. J. Bus. Res. 2018, 88, 428–436. [CrossRef]
- 25. Schlagwein, D.; Conboy, K.; Feller, J.; Leimeister, J.M.; Morgan, L. "Openness" with and without Information Technology: A framework and a brief history. J. Inf. Technol. 2017, 32, 297–305. [CrossRef]
- 26. Friesike, S.; Widenmayer, B.; Gassmann, O.; Schildhauer, T. Opening science: Towards an agenda of open science in academia and industry. J. Technol. Transf. 2015, 40, 581–601. [CrossRef]
- 27. Razavi, B. Fundamentals of Microelectronics, 2nd ed.; Wiley: Hoboken, NJ, USA, 2014.
- 28. Dieck-Assad, G.; Vázquez-Piñón, M. Simulation Practices for Electronics and Microelectronics Engineering, Volume 2, Chapter 5: CMOS Analog Microelectronics; Amazon eBooks: Monterrey, Mexico, 2013.
- 29. Gray, P.; Hurst, P.J.; Lewis, S.H.; Meyer, R. Analog Integrated Circuits, 5th ed.; John Wiley & Sons: Hoboken, NJ, USA, 2011.
- 30. Lehman, T.; Cassia, M. 1V Power Supply CMOS CascodeAmplifier. IEEE J. Solid State Circ. 2001, 36, 1082–1086. [CrossRef]
- 31. Lee, M.; Cho, S.; Lee, N.; Kim, J. New Radiation-Hardened Design of a CMOS Instrumentation Amplifier and its Tolerant Characteristic Analysis. *Electronics* **2020**, *9*, 388. [CrossRef]
- 32. Lim, A.; Tan, A.; Kong, Z.-H.; Ma, K. A Design Methodology and Analysis for Transformer-Based Class-E Power Amplifier. *Electronics* **2019**, *8*, 494. [CrossRef]
- Binkley, D.M. Tradeoffs and Optimization in Analog CMOS Design. In Proceedings of the 2007 14th International Conference on Mixed Design of Integrated Circuits and Systems, Date of Conference, Ciechocinek, Poland, 21–23 June 2007; pp. 47–60. [CrossRef]
- 34. González-Gómez, H.J.; Rodríguez-Vega, A. Micropower Op-Amps, Final Report for the TE-3034 Microelectronics Course Spring 2021 semester at Tecnológico de Monterrey, Mónterrey, México, June 2021, unpublished work.
- Lopez-Martin, A.J.; Ramirez-Angulo, J.; Carvajal, R.G.; Acosta, L. Micropower high current-drive class AB CMOS current-feedback operational amplifier. *Int. J. Circ. Theory Appl.* 2010, 39, 893–903. [CrossRef]
- 36. Pandit, S.; Mandal, C.; Patra, A. Nano-Scale CMOS Analog Circuits: Models and CAD Techniques for High-Level Design, 1st ed.; CRC Press: Boca Raton, FL, USA, 2014.
- 37. Rani, B.; Dhanoa, J.K. Designing of MOS Transistors in Two Stage OTA with Miller Compensation. *Int. J. Adv. Res. Electr. Electron. Instrum. Eng.* **2017**, *6*, 11, 8631–8637. [CrossRef]
- 38. Sadeqi, A.; Rahmani, J.; Habibifar, S. Design Method for Two-Stage CMOS Operational Amplifier Applying Load/Miller Capacitor Compensation. *Comput. Res. Prog. Appl. Sci. Eng. CRPASE: Trans. Electr. Electron. Comput. Eng.* **2020**, *6*, 153–162.
- 39. Qu, H. CMOS MEMS fabrication technologies and devices. *Micromachine* **2016**, *7*, 14. [CrossRef] [PubMed]
- 40. Peschot, A.; Qian, C.; Liu, T.J.K. Nanoelectromechanical Switches for Low-Power Digital Computing. *Micromachines* 2015, 6, 1046–1065. [CrossRef]
- 41. Huang, J.-Q.; Li, B.; Chen, W. A CMOS MEMS Humidity Sensor Enhanced by a Capacitive Coupling Structure. *Micromachines* **2016**, *7*, 74. [CrossRef] [PubMed]
- 42. Wang, J.; Yu, J. Multifunctional Platform with CMOS-Compatible Tungsten Microhotplate for Pirani, Temperature, and Gas Sensor. *Micromachines* 2015, *6*, 1597–1605. [CrossRef]
- Dieck-Assad, G.; Ávila-Ortega, A.; González Peña, O.I. Comparing Competency Assessment in Electronics Engineering Education with and without Industry Training Partner by Challenge-Based Learning oriented to Sustainable Development Goals. *Sustainability* 2021, 13, 10721. [CrossRef]
- 44. Dieck-Assad, G.; González Peña, O.I.; Rodríguez-Delgado, J.M. Evaluation of Emergency First Response's Competency in Undergraduate College Students: Enhancing Sustainable Medical Education in the Community for Work Occupational Safety. *Int. J. Environ. Res. Public Health* **2021**, *18*, 7814. [CrossRef]
- González-Peña, O.I.; Morán-Soto, G.; Rodríguez-Masegosa, R.; Rodríguez-Lara, B.M. Effects of a Thermal Inversion Experiment on STEM Students Learning and Application of Damped Harmonic Motion. *Sustainability* 2021, 13, 919. [CrossRef]

- 46. González-Peña, O.I.; Peña-Ortiz, M.O.; Morán-Soto, G. Is It a Good Idea for Chemistry and Sustainability Classes to Include Industry Visits as Learning Outside the Classroom? An Initial Perspective. *Sustainability* **2021**, *13*, 752. [CrossRef]
- 47. Parra Cordova, A.; Gonzalez Pena, O.I. Enhancing student engagement with a small-scale car that is motion-controlled through chemical kinetics and basic electronics. *J. Chem. Educ.* **2020**, *97*, 3707–3713. [CrossRef]